Author's response to reviews

Title: Anti-Hu Antibody associated Paraneoplastic Cerebellar Degeneration in Head and Neck Cancer

Authors:

Florian Huemer (florian.huemer@pmu.ac.at)
Thomas Melchardt (t.melchardt@salk.at)
Wolfgang Tränkenschuh (w.trenkenschuh@salk.at)
Daniel Neureiter (d.neureiter@salk.at)
Gerhard Moser (g.moser@salk.at)
Teresa Magnes (t.magnes@salk.at)
Lukas Weiss (lu.weiss@salk.at)
Alexander Schlattau (a.schlattau@salk.at)
Clemens Hufnagl (c.hufnagl@salk.at)
Gerda Ricken (gerda.ricken@meduniwien.ac.at)
Romana Höftberger (romana.hoeftberger@meduniwien.ac.at)
Richard Greil (r.greil@salk.at)
Alexander Egle (a.egle@salk.at)

Version: 2
Date: 16 February 2015

Author's response to reviews:

1) Line numbering was performed as requested

2) This article does not report the results of a controlled health-care intervention, subsequently there is no trial registration number

3) The CARE checklist was uploaded